\begin{table}%t4
\caption{\label{ahcnc_mintimes}Times of the primary and secondary minima in JD$_{\rm hel}$ (HJD- 2~400~000) for AH Cnc.}
%%\centerline

 {\small \begin{tabular}{llllll}
\hline \hline
HJD  & Meth. &  Ref. & HJD & Meth. &  Ref. \\
\hline
33~626.364	&	pg	&	1	&	51~939.1411	&	ccd	&	12	\\
34~421.307	&	pg	&	1	&	51~940.2221	&	ccd	&	12	\\
35~219.318	&	pg	&	1	&	51~940.2278	&	ccd	&	12	\\
36~656.352	&	pg	&	1	&	51~956.9876	&	ccd	&	13	\\
37~378.322	&	pg	&	2	&	51~957.1642	&	ccd	&	13	\\
37~699.300	&	pg 	&	2	&	51~958.0662	&	ccd	&	13	\\
38~820.7850	&	pe	&	3	&	51~958.2465	&	ccd	&	13	\\
39~964.300	&	pg	&	1	&	51~959.1478	&	ccd	&	13	\\
40~329.398	&	pg	&	1	&	52~314.1944	&	ccd	&	12	\\
40~570.9104	&	pe	&	4	&	52~314.2001	&	ccd	&	13	\\
40~678.3230	&	pe	&	5	&	52~315.0943	&	ccd	&	12	\\
41~396.332	&	pg	&	5	&	52~315.2771	&	ccd	&	12	\\
41~740.908	&	pg	&	6	&	52~995.2842	&	ccd	&	13	\\
41~742.8900	&	pe	&	6	&	52~996.3640	&	ccd	&	13	\\
41~752.8040	&	pe	&	6	& 52997.2642  &   ccd &   13  \\
41~797.6750	&	pe	&	6	&	53~001.2290	&	ccd	&	13	\\
41~815.6980	&	pe	&	6	&	53~004.1135	&	ccd	&	13	\\
42~537.3000	&	pe	&	6	&	53~004.2951	&	ccd	&	13	\\
43~163.554	&	pg	&	5	&	53~005.1948	&	ccd	&	13	\\
43~192.214	&	pg	&	5	&	53~005.3747	&	ccd	&	13	\\
43~256.371	&	pg	&	5	&	53~006.2758	&	ccd	&	13	\\
43~931.492	&	pg	&	5	&	53~007.1818	&	ccd	&	13	\\
44~015.288	&	pg	&	5	&	53~008.2609	&	ccd	&	13	\\
47~200.6990	&	ccd	&	7	&	53~009.3435	&	ccd	&	13	\\
47~200.8790	&	ccd	&	7	&	53~047.3702	&	ccd	&	14	\\
47~203.7620	&	ccd	&	7	&	53~083.0540	&	ccd	&	12	\\
47~203.9440	&	ccd	&	7	&	53~084.1348	&	ccd	&	12	\\
50~904.288	&	pg	&	8	&	53~426.3926	&	ccd	&	12	\\
51~159.3059	&	ccd	&	9	&	53~426.3931	&	ccd	&	14	\\
51~177.1469	&	ccd	&	9	&	53~437.7466	&	ccd	&	15	\\
51~177.3275	&	ccd	&	9	&	53~439.1876	&	ccd	&	12	\\
51~179.3103	&	ccd	&	9	&	53~442.7929	&	ccd	&	16	\\
51~229.7727	&	ccd	&	10	&	53~471.0869	&	ccd	&	12	\\
51~231.7490	&	ccd	&	10	&	53~489.2928	&	ccd	&	14	\\
51~245.8122	&	ccd	&	10	&	53~683.5814	&	ccd	&	14	\\
51~250.6798	&	ccd	&	10	&	53~750.6235	&	ccd	&	14	\\
51~273.0268	&	ccd	&	9	&	53~765.3982	&	ccd	&	17	\\
51~585.1847	&	ccd	&	11	&	54~060.9837	&	ccd	&	18	\\
\hline
\end{tabular}}
\medskip
1-~Kurochkin (\cite{Kurochkin1979}), 2-~Kurochkin (\cite{Kurochkin1965}), \mbox{3-~Eggen} (\cite{Eggen1967}), 4-~Millis (\cite{Millis1972}), 5-~Kurochkin (1970), 6-~Whelan et~al. (\cite{Whelan1979}), 7-~Estimated from light curves of Gilliland et~al. (\cite{Gilliland1991}), \mbox{8-~Diethelm} (\cite{Diethelm1998}), 9-~Youn et~al. (\cite{Yakut2003}), 10-~Blake (\cite{Blake2002}), 11-~Csizmadia et~al. (\cite{Csizmadia2002}), 12-~Qian et~al. (\cite{Qian2006}), 13-~Zhang et~al. (\cite{Zhang2005}), 14-~Krajci (\cite{Krajci2005}), 15-~Krajci (\cite{Krajci2006}), 16-~Nelson (\cite{Nelson2006}), 17-~Biro et~al. (\cite{Biro2006}), 18-~Nelson (\cite{Nelson2007}).
\end{table}